Author's response to reviews

Title: Relating circulating thyroid hormone concentrations to serum interleukins-6 and 10 in association with non-thyroidal illnesses including chronic renal insufficiency

Authors:

Hamdy A Zenah (abozenahamdy@hotmail.com)
Sabry A Shoeb (shoeb20@hotmail.com)
Alaa A Sabry (asabry20@yahoo.com)
Hesham A Ismail (ismail5@yahoo.com)

Version: 4 Date: 13 January 2008

Author's response to reviews:

Daer Sir:

Attached you will find our paper formatted as you asked.

Regarsd

Alaa